Author's response to reviews

Title: A Randomized Clinical Trial of a Peri-Operative Behavioral Intervention to Improve Physical Activity Adherence and Functional Outcomes following Total Knee Replacement

Authors:

Milagros C Rosal (Milagros.Rosal@umassmed.edu)
David Ayers (David.Ayers@umassmed.edu)
Wenjun Li (Wenjun.Li@umassmed.edu)
Carol Oatis (Oatis@arcadia.edu)
Amy Borg (Amy.Borg@umassmed.edu)
Hua Zheng (Hua.Zheng@umassmed.edu)
Patricia Franklin (Patricia.Franklin@umassmed.edu)

Version: 3 Date: 26 September 2011

Author's response to reviews: see over
September 26, 2011

Editor
BMC Musculoskeletal Disorders

MS: 1422075440583173

Dear Editor:

We are pleased with the news that our manuscript entitled “A randomized clinical trial of a peri-operative behavioral intervention to improve physical activity adherence and functional outcomes following total knee replacement,” will be published in *BMC Musculoskeletal Disorders*.

We have made the revisions you requested as outlined below:
- The full trial registration details have been added to the end of the abstract and have been included in the submission system.
- Figure 1 has been reduced to 1 page
- Author titles have been removed from the author list
- Countries have been added to the affiliations
- Methods and design has been changed to methods/design in the abstract
- Methods has been changed to methods/design in the main document
- Title and number has been removed from Figure 1

Please feel free to contact me with any additional questions or concerns at (508) 856-3173 or milagros.rosal@umassmed.edu.

Sincerely,

Milagros C. Rosal, Ph.D.
Corresponding Author